Author's response to reviews

Title: A investigation of cognitive 'branching' processes in major depression

Authors:

   Nicholas D Walsh (nickwalsh100@gmail.com)
   Marc L Seal (mseal@unimelb.edu.au)
   Steven CR Williams (s.williams@iop.kck.ac.uk)
   Mitul A Mehta (m.mehta@iop.kcl.ac.uk)

Version: 5  Date: 28 October 2009

Author's response to reviews:

Dear Editor,

Please find updated manuscript with formatting changes requested made.

Formatting changes requested
---------------------------------

Major revisions (we require the author to make these changes)

Authors- contributions- please indicate that all authors have read and approved the final version of the manuscript.

- Now done

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Please remove the figure and table placement text.

- Now done

Typography- Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

- Now done

Best wishes

Nicholas Walsh